Author's response to reviews

Title: Familial hypomagnesaemia with hypercalciuria and nephrocalcinosis (FHHNC): compound heterozygous mutation in the claudin 16 (CLDN16) gene.

Authors:

Geeta Hampson (geeta.hampson@kcl.ac.uk)
Martin A Konrad (konradma@uni-muenster.de)
John Scoble (john.scoble@gstt.nhs.uk)

Version: 4 Date: 2 July 2008

Author's response to reviews:

Response to reviewer 3:
1. We have shortened the discussion.
2. The spelling errors (FHNNC instead of FHHNC) have been corrected throughout the manuscript.
3. The list of authors has been written exactly as they are in the submission system.
4. An acknowledgement section has been added